Author's response to reviews

Title: Hypoxia induces differential translation of enolase/MBP-1

Authors:

Kara C Sedoris (kjcoll01@louisville.edu)
Shelia D Thomas (sdthom02@louisville.edu)
Donald M Miller (donaldmi@ulh.org)

Version: 7 Date: 2 April 2010

Author's response to reviews: see over
Dear Sir/Madame;

Thank you for your consideration of our manuscript **HYPOXIA INDUCES DIFFERENTIAL TRANSLATION OF ENOLASE/MBP-1**. Only a few final changes have been made 1) renumbering of figures 1A, 1B in the results and figure legend and 2) renumbering of figure 2A and 2B in the results. Thank you kindly for all of your help and suggestions. We will definitely consider Biomed Central for future publications.

Sincerely,

Kara Sedoris